Author’s response to reviews

Title: The state of research on cyberattacks against hospitals and available best practice recommendations: A scoping review

Authors:

Salem Argaw (sta47@cornell.edu)
Nefti-Eboni Bempong (Nefti-eboni.bempong@unige.ch)
Bruce Eshaya-Chauvin (Beshayachauvin@gmail.com)
Antoine Flahault (Antoine.Flahault@unige.ch)

Version: 2 Date: 28 Nov 2018

Author’s response to reviews:

Technical Comments:

-Please change the heading "Methodology" to "Methods"
Addressed.

-All figure titles/legends should be listed and placed at the end of the main manuscript, after the References, and not within any of the figure files.
Addressed.

Editor Comments:

Reviewer #2: Thank you for having me to review the revision of this manuscript. The authors have addressed all the reviewers' comments and made relevant changes. I do not have further feedback.

Thank you for your contribution.

Editorial Comments:

Methods: the statement on page--2 lines 62--63 "Data from articles were then extracted and analyzed, using descriptive and narrative method"
The DESCRIPTIVE ANALYSIS requires few lines either in the discussion section or Limitation section to defend about the approach.

It is necessary to remove the cluttering of this term where there is NO Descriptive Analysis performed statistically.

Thank you for pointing this out. We have taken your comment into consideration and removed the phrase as it seemed to be misleading.